Author's response to reviews

Title: Achieving sustained virologic response in hepatitis C: A review of the clinical, economic and quality of life benefits

Authors:

Jayne Smith-Palmer (smith-palmer@ossianconsulting.com)
Karin Cerri (CERRI@its.jnj.com)
William Valentine (valentine@ossianconsulting.com)

Version: 2 Date: 20 August 2014

Author's response to reviews: see over
Dear Dr Ramos,

Revisions required for a review article “Achieving sustained virologic response in hepatitis C: A review of the clinical, economic and quality of life benefits” for submission to *BMC Infectious Diseases*

We have made revisions to our manuscript as requested and would like to submit our above titled article for consideration in *BMC Infectious Diseases*. In line with requested changes, we have made the following revisions to the manuscript submitted on 19 August 2014:

- Line numbers have been added to the Word version of the manuscript and all page breaks have been removed
- A competing interests section has been added outlining the source of funding for the manuscript and competing interests for all of the authors
- A section outlining authors’ contributions has been added
- Email addresses for all authors have been added to the title page

We look forward to hearing from *BMC Infectious Diseases*.

Yours faithfully

Jayne Smith-Palmer